Author's response to reviews

Title: Pheochromocytoma presenting with arterial and intracardiac thrombus in a 47 year old woman: a case report

Authors:

Runhua Hou (runhua_hou@urmc.rochester.edu)
Ann M Leathersich (aleathersich@path.wustl.edu)
Brenda T Ruud (alwaysruud2@hotmail.com)

Version: 3 Date: 6 December 2010

Author's response to reviews:

Dear Editor,

Thank you very much for the reviews for our paper “Pheochromocytoma presenting with arterial and intracardiac thrombus in a 47 year old woman: a case report” by Hou et al. We are pleased to have this opportunity to address concerns raised by the editor. We asked Barbara Morabito, a native English speaking colleague to copy-edit our paper. Please see all changes which are highlighted in the revised text.

We hope that you will agree with these changes and reconsider our revised manuscript for publication in JMCR. We are happy to discuss with you further if there are any other questions.

Thank you very much. We look forward to hearing from you soon.

Sincerely yours,

Runhua Hou